Author’s response to reviews

Title: Managing Obstetric Emergencies and Trauma (MOET) Structured skills training in Armenia, utilising models and reality based scenarios

Authors:

Richard B Johanson (richard@kogs.freeserve.co.uk)
Vijay Menon (linda@lucking5.freeserve.co.uk)
Ethel Burns (eburns@brookes.ac.uk)
Eduard Kargramanya (linda@lucking5.freeserve.co.uk)
Vardges Osipov (linda@lucking5.freeserve.co.uk)
Musheg Israelyan (linda@lucking5.freeserve.co.uk)
Karine Sargsyan (linda@lucking5.freeserve.co.uk)
Sarah Dobson (linda@lucking5.freeserve.co.uk)
Peter Jones (p.w.jones@maths.keele.ac.uk)

Version: 5 Date: 16 May 2002

In this latest version of our manuscript (MOET Armenia paper (resubmitted) 17-05-02) the title page has been revised to change the corresponding author from Professor Richard Johanson (recently deceased) to Mr Vijay Menon. Details of author contributorship have also been included after the Conclusion section.